Author's response to reviews

Title: Immediate processing of erotic stimuli in paedophilia and controls: a case control study

Authors:

Benedikt Habermeyer (benedikt.habermeyer@puk.zh.ch)
Fabrizio Esposito (fabrizio.esposito@unina.it)
Nadja Händel (Nadja.Haendel@upkbs.ch)
Patrick Lemoine (Patrick.Lemoine@upkbs.ch)
Markus Klarhöfer (Markus.Klarhoefer@unibas.ch)
Ralph Mager (ralph.mager@upkbs.ch)
Volker Dittamnn (volker.dittmann@upkbs.ch)
Erich Seifritz (seifritz@bli.uzh.ch)
Marc Graf (Marc.Graf@upkbs.ch)

Version: 2 Date: 29 August 2012

Author's response to reviews:

Catherine Olino
Journal Editorial Office
BioMed Central

Dear Catherine Olino

I removed the pictures from the manuscript as you suggested (i.e. I replaced Figure 1). Furthermore I added in the Method section more information about obtaining consent and the age of the study subjects.

I hope that you will now be able to start the review process

Please do not hesitate to contact me again if you would need further changes.

Best wishes

Benedikt Habermeyer